S Ш 2 PA U Z 0 ×

EAST-WEST CENTER WORKING PAPERS

EAST-WEST CENTER

The U.S. Congress established the East-West Center in 1960 to foster mutual understanding and cooperation among the governments and peoples of the Asia Pacific region including the United States. Funding for the Center comes from the U.S. government with additional support provided by private agencies, individuals, corporations, and Asian and Pacific governments.

East-West Center Working Papers are circulated for comment and to inform interested colleagues about work in progress at the Center.

For more information about the Center or to order publications, contact:

Publication Sales Office East-West Center 1601 East-West Road Honolulu, Hawaii 96848-1601

Telephone: 808-944-7145 Facsimile: 808-944-7376 Email: ewcbooks@EastWestCenter.org Website: www.EastWestCenter.org EAST-WEST CENTER

Economics Series

No. 6, September 2000

The Impact of Imported and Domestic Technologies on Productivity: Evidence from Indian Manufacturing Firms

Rana Hasan

Rana Hasan is a Fellow at the East-West Center. This paper is part of his research on the effects of market-oriented economic reforms on productivity growth, labor markets, and inequality.

East-West Center Working Papers: Economics Series is an unreviewed and unedited prepublication series reporting on research in progress. The views expressed are those of the author and not necessarily those of the Center. Please direct orders and requests to the East-West Center's Publication Sales Office. The price for Working Papers is \$3.00 each plus postage. For surface mail, add \$3.00 for the first title plus \$0.75 for each additional title or copy sent in the same shipment. For airmail within the U.S. and its territories, add \$4.00 for the first title plus \$0.75 for each additional title or copy in the same shipment. For airmail elsewhere, add \$7.00 for the first title plus \$4.00 for each additional title or copy in the same shipment.

ABSTRACT

Proponents of trade liberalization in developing countries often argue that one of its most important benefits is that it enables firms in developing countries to access the international knowledge base by importing technology in both disembodied form (i.e., as technological knowhow) as well as embodied form (i.e., embodied in imported capital goods). Opponents of trade liberalization argue otherwise. In addition to doubting that there are significant gains to be had from utilizing foreign technologies in developing country contexts, they believe that imports of technology dampen local efforts at developing new technology with negative consequences for local capabilities and long-run growth prospects.

This paper utilizes panel data on a sample of Indian manufacturing firms for the years 1977-87 to examine these views. Production function estimates reveal that imported technologies, especially those of disembodied nature and obtained through contractual arrangements with foreign firms, impact productivity positively and significantly. Firms' own R&D efforts, on the other hand, are not very productive. Finally, while domestically produced capital goods impact productivity positively and significantly, their impact appears to stem from the technological know-how imported by domestic producers of capital goods.

Although these findings support the optimism of liberalizers that foreign technologies represent an important opportunity for productivity enhancement for developing country firms, the estimates of this paper also lend support to the notion that a liberal import policy will dampen local efforts at developing new technologies. More specifically, the estimates reveal that firms do not need to undertake significant R&D efforts to utilize imported technologies effectively. Thus, taken together the results suggest that while firms in India's recently liberalized economic environment will be able to raise their productivity by importing greater amounts of foreign technologies, they will also have less incentives to carry out their own R&D. To the extent that local efforts at R&D are a "good" to be encouraged, the challenge for public policy will be to devise policy tools that are able to encourage local R&D, but not through a trade policy which blocks an important and direct channel by which firms can raise productivity.

ACKNOWLEDGEMENTS

I am extremely grateful to the Institute for Studies in Industrial Development of New Delhi, India for allowing me to use the firm-level data utilized in this study. I have received many helpful comments from Roger Betancourt, Brian Fikkert, Charles Hulten, Lane Koenig, and Mayank Raturi. However, I alone am responsible for any errors.

1. Introduction

There has been a resurgence of interest in the role of international trade in promoting economic growth and productivity in recent times. In particular, the emphasis of new growth theories on the growth and productivity enhancing role of investments in technical knowledge by profit seeking firms has led a number of researchers to contemplate how trade may influence these investment decisions. At one level, trade is believed to affect firms' incentives to invest in technology via its impact on market structure and competition.¹ At another, it is believed to a country and its firms. Indeed, as Grossman and Helpman (1991) write, "perhaps the most important benefit to a country from participating in the international economy might be the access that such integration affords to the knowledge base in existence in the world at large" (pg. 238).

There are a number of ways in which a country may access the international knowledge base. Perhaps the most direct and important of these are the import of physical inputs which embody new and/or superior foreign technology (particularly capital goods) and the import of designs, blueprints, technical assistance, and training from foreign firms. The latter represent imports of disembodied technology as they primarily involve the transfer of technical knowledge and skills.²

Unfortunately not too much is known about the effects of these technology/knowledge imports on firms' productivity and their interactions with domestic efforts at understanding and generating technology. In fact, in contrast to the optimistic picture painted by the statement of Grossman and Helpman above, there are doubts in the minds of economists and policy makers in some less developed countries (LDCs), particularly India, about the benefits of foreign technology for the growth prospects of LDCs. A commonly held view has been that the import of disembodied technology takes place at the behest of foreign firms looking for quick profits and that the technology transferred does not represent any significant gain in performance or productivity for the Indian firms.³

¹ It may be noted that contrary to popular perceptions, the new framework does not provide an unambiguous theoretical underpinning to the notion that opening to trade increases the incentives of less developed country (LDC) firms to invest in innovations (see for example, Rodrik 1992, Young 1991).

² There are a number of other channels as well. For example, case study evidence suggests that export activities by firms in LDCs lead to improvements in their technical efficiency as international buyers transmit technical assistance and knowledge to them. While a number of econometric studies purport to show the same, see Clerides, Lach, and Tybout (1996) for a critique of these studies. Spillovers of international knowledge represent yet another mechanism of drawing upon the international knowledge base. In principle, such spillovers do not have to be related to trade. In practice, however, even these are likely to depend on trade (Coe and Helpman, 1995).

³ For example, case studies by Scott-Kemmis and Bell (1988) and Desai (1988) examining the contractual purchase of foreign technology by Indian firms suggest that the importing Indian firms do not get enough technological know-how through the contracts. Although this evidence has been interpreted differently by various researchers, many have taken the case study evidence to be indicative of low returns from the import of technology. Indeed, even Arora (1996) who clearly believes technology transfer to be a potentially important source of technology and focuses on policies that would improve the efficiency of technology transfer is prompted to write that the "ability of Indian firms [engaged in arm's length import of technology] to assimilate, utilize, and improve

Another view holds that even if the imports of technology (embodied or disembodied) are productivity improving, they lead to a dampening of domestic R&D activity as local firms purchase technology (embodied or disembodied) from foreign ones. If there are significant externalities from domestic R&D efforts, then a liberal technology import policy could be detrimental for an LDC's long-term growth prospects (Kumar 1990). However, whether imports of technology substitute for local efforts at technology generation or not is the subject of much debate. Indeed, some economists have argued that imports of technology <u>spur</u> domestic R&D activity because of the need to adapt and utilize effectively imported technology (see Katrak 1997 for a recent review of the debate).

This paper uses panel data from 1976-77 to 1986-87 for a sample of Indian manufacturing firms to determine the extent to which the productivity of Indian firms has been affected by their investments in imported and domestic sources of technology and the nature of these technology inputs' interactions in production, thus shedding some light on the views above. Production functions augmented with various technology inputs are estimated using a variety of econometric specifications. In particular, the technology inputs include not only firms' stocks of R&D, but also their stocks of imported disembodied technology following Basant and Fikkert (1996), Raut (1995), and Ferrantino (1992). However, because new technologies are widely believed to be embodied in the capital goods used by firms, this paper also includes measures of embodied technical change as additional determinants of firms' productivity.

The inclusion of technology embodied in capital goods as a possible source of technological change is an important one. For instance, one interpretation of cross-country growth regressions that find investments in capital goods, particularly of the imported variety, to have a strong and robust impact on economic growth (De Long and Summers 1991, 1993; Lee 1995), is that new capital goods embody superior technology. Unfortunately, the results of cross-country growth regressions, are not easy to interpret as they tend to be prey to a number of econometric problems and aggregation biases which may be alleviated by using micro-level panel data. By using data on the sample firms' investments in domestic and imported capital goods this paper attempts to provide direct, micro-based evidence on the impact of technology embodied in capital goods on productivity.

The paper is organized as follows. Section 2 discusses the conceptual framework and estimation strategy used in this paper. Section 3 presents a description of the data set and the variables used in estimation. Section 4 details the results from the empirical analysis and compares these with the earlier literature. Finally, section 5 concludes.

2. Conceptual Framework and Estimation Issues

2.1 Conceptual Framework

A firm is assumed to produce output Q using capital K, labor L, and materials M, subject

the technology is limited."

to a state of technology S, in accordance with a production function:

$$Q' F(K,L,M,S).$$
(1)

A firm's state of technology is assumed to depend not only on the technical knowledge possessed by it but also the technology embodied in the capital goods it uses. Improvements in a firm's state of technology represent increases in its productivity or technical efficiency.

Most studies in this strand of the literature have been conducted for developed countries and have generally focused on firms' own R&D as the sole determinant of technical knowledge.⁴ While this may be realistic in the case of developed countries, it is not likely to be a good one in the context of an LDC like India. This is because in addition to conducting their own R&D, Indian firms' routinely acquire technical knowledge through the import of disembodied technology from foreign firms. These imports take place through technology transfer agreements with foreign firms which supply designs, blueprints, and technical assistance to the contracting domestic firm. Thus the framework adopted in this paper considers there to be two distinct sources of technical knowledge: own R&D and the purchase of disembodied technology from foreign firms.

As far as technology embodied in capital goods is concerned it is quite difficult, if not impossible, to measure how much technology is embodied in a particular piece of equipment. A simplification made by many researchers is to assume that recently purchased capital goods embody newer, more productive vintages of technology than older capital goods (for example, Bahk and Gort 1993; Baily, Hulten, and Campbell 1992; Bregman, Fuss, and Regev 1991).⁵ In this paper, these embodiment effects are captured by using the shares of the total capital stock made up of recent investments in imported capital goods and recent investments in domestic capital goods, an approach which minimizes multicollinearity problems in estimation and is along the lines of

⁴ Some studies also incorporate a measure of the R&D preformed by other firms belonging to the same industry in order to capture the importance of R&D spillovers. Two recent studies have examined similar issues for India. See Basant and Fikkert (1996) and Raut (1995).

⁵ A couple of factors would seem to suggest that in the Indian context, this assumption is more likely to hold for recently imported capital goods than recently produced domestic capital goods. First, India's trade and industrial policies restricted the imports of capital goods on account of a desire to protect India's capital goods producing industries and to conserve foreign exchange (Ahluwalia 1994). A complex licensing procedure ensured that the import of a capital goods actually imported would tend to embody newer technologies than those produced domestically. Second, it is doubtful that Indian capital goods and thus produce capital goods that embody significantly new technologies (Lall 1987). However, it has to be factored in that Indian capital goods producers would be able to tap into the superior technical knowledge required to produce technologically advanced capital goods by entering into technology transfer agreements with foreign capital goods producers. Thus, domestically produced capital goods could well embody productivity improving technological advances from abroad.

Bregman, Fuss, and Regev (1991) and Berndt and Morrison (1995).⁶

In summary, Indian firms' state of technology is assumed to depend on their R&D, their purchases of disembodied technology from foreign firms, and the shares of their capital that are composed of recent investments in imported and domestic capital goods:

$$S' f(R, T, RI^{M}/K, RI^{D}/K), \qquad (2)$$

where R and T represent the stocks of technical knowledge generated by a firm's own R&D expenditures and purchases of disembodied technology from foreigners, respectively, and RI^M/K and RI^D/K represent firms' shares of new imported and domestic capital. Firms with higher levels of R and T and, therefore, a higher state of technology, should then be the more productive ones. Similarly, if new capital goods embody advances in technology and these advances improve productivity, then the shares of capital that are of recent vintage should impact output positively in a production.

In order to estimate the four-input production function in equation (1), two functional forms are used: the Cobb-Douglas and Translog, which can be viewed as first and second order approximations to any production function, respectively. The Cobb-Douglas takes the form:

$$\ln Q_{it} \stackrel{\prime}{} A_t \stackrel{\prime}{}_{j'1} \frac{4}{\gamma_j} \log S_{jit} \stackrel{\prime}{}_{k'1} \frac{3}{\beta_k} \log X_{kit} \stackrel{\prime}{}_{k'i} \varepsilon_{it}, \qquad (3)$$

where Q is as previously defined, $S=(S_1, S_2, S_3, S_4)=(R, T, RI^M/K, RI^D/K)$, $X=(X_1, X_2, X_3)=(K, L, M)$, A represents a time-specific intercept term that captures exogenous, economy-wide shifts in productivity, ε denotes an error term, and i and t index firm and time, respectively.

Similarly, the Translog takes the form:

$$\ln Q_{it} \stackrel{\prime}{} A_t \stackrel{4}{\aleph_{j}}_{j'} \frac{\gamma_j \log S_{jit}}{\gamma_j} \stackrel{8}{\aleph_j}_{k'} \frac{\beta_k \log X_{kit}}{\gamma_k} \stackrel{8}{\aleph} (1/2) \stackrel{3}{\circledast_{k'}} \frac{\beta_{kl} \log X_{kit}}{\gamma_k} \stackrel{8}{\aleph} \frac{\beta_{kl} \log X_{kit}}{\gamma_k} \frac{\varphi_{kl}}{\varphi_{kl}} \frac{\varphi_{kl}}$$

where all of the variables are as previously defined. Note that while a firm's state of technology is

⁶ One alternative to the share approach is to measure the embodiment effect by including the actual levels of recently purchased capital goods, imported and domestic (RI^M and RI^D). This, however, would exacerbate multicollinearity problems in estimation on account of the inclusion of total K in the production function. K includes all investments in fixed assets and would thus include recent investments in capital goods, domestic and imported. This double counting is likely to cause problems of multi-collinearity in the regressions. Indeed, the partial correlations between K and RI^D and K and RI^M (taking into account firm effects) are 0.67 and 0.57, respectively. These are substantially larger than the corresponding partial correlations between K and RI^D/K and K and RI^M/K, 0.25 and 0.33, respectively, suggesting that the usage of the shares suffer less from multi-collinearity problems.

assumed to be additively separable throughout, in some of the empirical analysis below the various technology inputs are allowed to interact with one another, thereby allowing the inputs to be substitutes or complements for one another in production in the sense that two inputs are substitutes (complements) for one another if an increase in the usage of one leads to a decline (increase) in the output elasticity of the other. Finally, because the various technology inputs can take zero values across observations, a 1 has been added to each of the technology inputs as in Raut (1995) in order to avoid the problem of having to take a log over a zero.

2.2 Estimation Issues

Estimation of the production functions above requires that some assumptions be made about the error term. In view of the panel nature of the data, the error term is assumed to be composed of two terms:

$$\varepsilon_{it} \, \, ' \, \, u_i \, \, \% \, v_{it}. \tag{5}$$

The first term, u_i , corresponds to firms' unobserved time-invariant technical efficiency and the second term, v_{it} , captures the residual variation in output. If u_i is not correlated with the regressors, then OLS is consistent. However, it is not efficient as it does not take into account the fact that part of the error term is constant for each firm over time. A gain in efficiency may be made by using the random effects estimator.

The assumption that u_i is not correlated with the regressors is a rather strong one. In particular, while a firm's level of time-invariant technical efficiency may be unobservable to the econometrician it is very likely to be known to the firm. If this is the case, then firms may be expected to make input choices on the basis of their technical efficiency. In other words, u will be correlated with the regressors and treating it as part of the error term, as is done in the random effects model, will introduce an omitted variable bias. One way to deal with this is to transform the data in order to sweep away the u_i. This may be done by expressing each variable in terms deviations around its firm-specific (within-group) mean value (for example, x_{it} - ' x_{it}/T), as in the so-called fixed effects model.⁷ Alternatively, it is possible to express each variable in differenced form (for example, x_{it} - x_{it-1}) and thereby rid the regression equation of the u_i . Which particular procedure is used usually depends on whether the other component of the error term, v_{it}, is assumed to be endogenous or not. In particular, if v_{it} is assumed to be correlated with the regressors then an instrumental variables estimator can yield consistent estimates of the regressors. In many applications, finding valid instruments is easier for the first differenced estimator as compared to the within-group estimator. This is because, typically, the only valid instruments available for panel data on micro-level variables are lagged values of the regressors themselves. Since the error term of the within-group or fixed effects model contains shocks from every period in the panel (ε_{it} -' $\varepsilon_{it}/T = v_{it} - v_{it}/T$), it will be correlated with lagged values of the regressors under the assumption

⁷ A Hausman specification test can be used to determine whether the fixed effects or random effects model is appropriate.

of endogenous v_{it} , rendering these invalid as instruments. This is not a problem with the first differenced transformation, however. Since the error term for that model is $(\epsilon_{it}-\epsilon_{it-1})=(v_{it}-v_{it-1})$, a lagged regressor, say x_{it-2} , will not be correlated with either v_{it} or v_{it-1} as long as the v_{it} 's themselves are not autocorrelated.⁸

3. Data and Variable Construction

The data set used in this paper consists of the annual reports of 286 public limited firms, defined as private corporations with more than 50 shareholders, covering the years 1975-76 to 1986-87.⁹ The firms may be placed into 8 manufacturing industries at the approximately two-digit level, spanning the broad range of the manufacturing sector: 5 percent of the firms belong to food processing; 17 percent to textiles; 11 percent to light manufacturing (which is taken here to include firms producing paper, cement and rubber and plastic goods); 8 percent to metal products; 20 percent to chemicals and 8 percent to pharmaceuticals; 9 percent to transportation equipment; 12 percent from electrical machinery; and 10 percent from non-electrical machinery.

The variables required for the estimation of the production functions in equations (3) and (4) are firm-level measures of output, capital stock, labor, material inputs, stocks of R&D and imported technology, and measures of technology embodied in imported and domestic capital goods. These are described below.

Output (Q): Each firm's value of production for any given year was deflated to constant 1985-86 rupees using the wholesale output price deflators for the firm's industry.¹⁰ These deflators were obtained from the <u>Index Numbers of Wholesale Prices of India</u> contained in Chandhok and the Policy Group (1991).

Labor (L): Labor units were derived by dividing each firm's total payments for wages and salaries by industry-specific wages for each year. The industry-specific wage data was computed by dividing each industry's "total emoluments to employees" by the "total number of labor hours worked" using the industry-level data reported in the <u>Annual Survey of Industries of India</u>.

Materials (M): Each firm's expenditures on materials and intermediate inputs were deflated by an industry-specific materials price deflator. India's input-output matrix was used to weight output price indices in order to construct an industry-specific, materials price deflator. The source of the output price indices was once again the <u>Index Numbers of Wholesale Prices of India</u> contained in

⁸ Even if the v_{it} 's were autocorrelated following, for example, an AR1 process, a suitable transformation of the original model along with a sufficiently lagged regressor can still serve as a valid instrument as is shown in Appendix 2 below.

⁹ I am extremely grateful to the Institute for Studies in Industrial Development of New Delhi, India for allowing me to use the firm-level data utilized in this study.

¹⁰ The industry-wide deflators are based on the lowest level of aggregation possible. For most variables this is the three-digit level (the firms can be assigned to roughly 29 industries at the three-digit level).

Chandhok and the Policy Group.

Capital (K): Firms' capital stocks were constructed using the perpetual inventory method:

$$K_{it} \stackrel{\prime}{} I_{it} \,\% \, K_{it\&1} \, \varPhi 1 \, \& \delta], \tag{6}$$

where I_t denotes real investment, calculated as the difference in firms' total gross fixed assets deflated by the wholesale price index for investment goods, P_{kt} , and δ denoted the rate of depreciation, assumed to be six percent. Details of the procedure, including the construction of starting values of the capital stock are given in Appendix 1.

(R) R&D Capital: Firms' expenditures on R&D activities were converted into stocks of R&D by using the perpetual inventory method to aggregate firms' annual real investments in R&D.¹¹ Following previous research, R&D was assumed to affect output with a one year lag and decay at the rate of 15 percent per annum..¹² The evolution of the stock of R&D may be expressed as:

$$R_{it} ' RD_{it\&1} \% R_{it\&1} @1\&\delta], (7)$$

where R is the stock of R&D generated knowledge and RD are the annual, real expenditures on R&D activities.

Several steps had to be undertaken before the above equation could be applied to the data. First, the annual expenditures on R&D activities had to be converted into real values. For the annual expenditures on R&D, this was accomplished by using the average of the wage and capital goods deflators as the R&D deflator. Using these deflators, the real expenditures, RD, were obtained.

The next step was to come up with starting values for R. If a firm had incurred no expenditures on R&D in the first year for data, then it was assumed that the firm did not conduct R&D in any of the previous years. This assumption is not as strong as it may seem; most firms in India did not conduct R&D prior to 1975-76, the first year for which the firm-level R&D data is

¹¹ A cross-checking of the R&D numbers from the firms' annual reports with those contained in the <u>Compendium on In-House R&D Centers</u> (Department of Science and Technology) revealed that a number of firms failed to itemize their R&D expenditures in their annual reports. For such firms, the R&D expenditures reported to the Department of Science and Technology were used as in Fikkert (1994).

¹² Experimentation with alternative assumptions about the lag with which R&D (and imports of disembodied technology) impacts productivity and depreciation rates did not alter results very much. In particular, while a one year lag and depreciation rate of 15 percent may be appropriate for data from developed countries, it is possible that in a more protected and less competitive market such as India's, technical knowledge becomes obsolete at a lower rate, thereby implying a lower depreciation rate. Similarly, the lower skills of the labor force may mean that it takes longer for investments in technical knowledge to impact output. However, experimentation with lower depreciation rates (for example, 6 percent) and longer impact lags (for example, two years) yielded parameter estimates very similar to those based on a 15 percent rate of depreciation and impact lag of one year.

available. For firms conducting R&D in 1975-76, it was necessary to determine the number of years for which firms had been doing R&D prior to 1975-76 and also the growth in real R&D expenditures over the period. The Department of Science and Technology's <u>Research and Development Statistics</u> can be used to estimate these. Computations reveal that among the Indian firms which had been recognized by the Department of Science and Technology in 1975-76 as operating R&D units, the average age of the R&D unit was about 5 years and that real R&D expenditures per firm had grown by about 5 percent per annum.

Assuming a rate of deprecation of 15 percent and a one period lag in the impact of R&D expenditures on output, the stock of R&D in such firms in 1975-76 is:

$$R_{i76} \, \, ' \, RD_{i75} \, \% (1\&\delta) \, \mathcal{C}D_{i74} \, \% \dots \, \% (1\&\delta)^4 \, \mathcal{C}D_{i71}, \tag{8}$$

where, the year 1975-76 is written as 76 for notational convenience, etc. Further, since real R&D expenditures per firm had grown by about 5 percent per annum, the above may be rewritten as:

$$R_{i76} \stackrel{\prime}{=} \frac{RD_{i76}}{1\%0.05} \stackrel{4}{=} \frac{180.15}{1\%0.05} \stackrel{s}{=} .$$
(9)

For all subsequent years, equation (8) was applied.

(T) Imported Disembodied Technology Capital: Stocks of imported disembodied technology were constructed by applying the perpetual inventory method to firms' annual real imports of disembodied technology obtained through technology transfer agreements with foreign firms. As in the case of R&D stocks, the import of disembodied technology (TP) was assumed to impact output with a one year lag and decay at the rate of 15 percent per annum. The stock of imported disembodied technology T is assumed to evolve as:

$$T_{it} ' TP_{it\&1} \% T_{it\&1} @1\&\delta].$$
(10)

The annual expenditures on disembodied technology imports include payments to foreigners for technical assistance and consulting plus lump sum and royalty payments for the purchase of technology through licensing agreements between Indian and foreign companies. In order to deflate these technology import expenditures the procedure used was to adjust the R&D deflator for U.S. manufacturing industry for changes in the Dollar-Rupee exchange rate. The R&D deflator for U.S. industry was considered appropriate since U.S. firms constitute the largest source of transfers of technology to Indian firms.

To compute the starting values for T for each firm it is necessary to obtain information on firms' histories of disembodied technology import. The <u>Directory of Foreign Collaborations in India</u> lists the years in which Indian firms entered into technology based licensing agreements with foreign

firms. This publication was used to determine the years in which the sample firms entered into such licensing agreements from 1965-75 and the flows of technology based on these licensing agreements were assumed to last for four years on the basis of Kapur (1983). To derive numbers that would represent the payments for these technology flows for the identified years, the initial year's sales of the firms with the pre-sample technology imports were multiplied with the average three-digit disembodied technology import to sales ratio for the initial year. Finally, all the expenditures on disembodied technology import, including the pre-sample expenditures just described above, were deflated and used in equation (11) to get the stocks of imported disembodied technology, T.

(RI^M/K) Technology Embodied in Imported Capital Goods: The annual reports of firms also detail investments in imported capital goods and the book values of plant and equipment. The expenditures on imported capital goods were deflated by the unit value index of imported capital goods available in the <u>Statistical Abstract of India</u> to arrive at real investments in imported capital goods (I^M). Real total investments in plant and equipment (I^P) were derived by differencing the book values of plant and equipment and then deflating the resulting figures by the wholesale price index of capital goods.¹³

The stock of recent investments in imported capital goods, RI^M, may be expressed as:

$$RI_{it}^{M} \stackrel{'}{}_{j_{s'0}} I_{it\&s}^{M} (2 \&\delta)^{s}.$$

$$(11)$$

The past flows were discounted using depreciation rates of six percent and a value of T=4 was chosen as in Bregman, Fuss, and Regev.¹⁴ The share of total capital stock made up of recent investments in imported capital goods is then RI^M/K .

(RI^D/K) Technology Embodied in Domestic Capital Goods: The stock of recent investments in domestic capital goods can be constructed as:

$$RI_{it}^{D} \, ' \, \sum_{s' \, 0}^{T} I_{it\&s}^{P} (1\&\delta)^{s} \& RI_{it}^{M},$$
(12)

where RI^{M} is as defined in equation (12) and I^{P} , representing total investment in plant and equipment, is composed of investments in domestic and imported capital goods in that period. The share of total capital stock made up of recent investments in domestic capital goods is then RI^{D}/K .

¹³ Recall that (total) capital, K, is made up of investments in fixed assets. Fixed assets can be sub-divided into either (i) land and buildings, or (ii) plant and equipment. It is the latter which may be either imported or domestically produced.

¹⁴ The results were not sensitive to an alternative rate of fifteen percent for depreciation. The results were also not sensitive to a choice of T=5 or T=3.

In order to use as many years of data as possible it is necessary to estimate pre-sample investments in imported capital goods and total plant and equipment. First, the average investments in imported capital goods and total plant and equipment from 1976-77 to 1979-80 were computed for each firm.¹⁵ An average was taken instead of the first year's investments since it is likely to be a more representative indicator of a firm's pattern of investments. Next, using the national accounts statistics presented in Chandhok and the Policy Group (1990) and <u>Statistical Abstract of India</u>, the aggregate growth rates of real investments in total plant and equipment and imported capital goods were computed for the 1972-73 to 1977-78 period.¹⁶ These were found to be 6.4 percent and 1.2 percent per annum on average, respectively. Finally, these aggregate growth rates were used to compute firms' pre-sample investment levels of imported capital goods and total plant and equipment using each firms' average investments in imported capital goods and total plant and equipment (computed in the first step) as a base.

Table 1a-c describes various statistics relating to the variables used in estimation by twodigit industry. As can be seen from examining the columns relating to the technology inputs, the various industries can vary significantly in terms of the propensity with which firms use these, especially technical knowledge. Thus, while only 35 percent of firms in textiles conducted any R&D, 88 percent of firms did so in electrical machinery (Table 1b). Similarly, while only 13 percent of the firms in the food processing industry imported disembodied technology 93 percent of firms did so in non-electrical machinery. These numbers, along with those on the intensity of various technology stocks with respect to output (and detailed in Table 1c) suggest that technology opportunities vary by industry (and that even within an industry firms will vary in the extent to which they are willing to exploit these opportunities) and that it is important that production functions be estimated over firms belonging to technologically homogenous groupings.

While the two digit industry level presents itself as a basis to group firms over, the relatively small number of firms in many of the two digit industries can leave econometric estimates susceptible to small sample biases. Thus, in what follows, estimation is performed over firms which are pooled over more aggregated groupings that are likely to be technologically homogeneous. In particular, following Cueno and Mairesse (1984), Griliches and Mairesse (1984), and Basant and Fikkert (1996), the sample firms are divided into either a scientific group - consisting of 113 firms belonging to the electrical, chemical, and pharmaceutical industries - or a non-scientific group - consisting of 173 firms belonging to the remaining industries. Although such a grouping of firms may be appropriate for developed countries where technological opportunities appear greatest in the electrical, chemical, and pharmaceutical industries of the less sophisticated Indian context is unclear. For example, on the basis of the intensities of the technical knowledge stocks presented in Table 1c, it would appear that in addition to the electrical machinery also invest

¹⁵ Recall that because domestic investments are derived by differencing the book values of plant and machinery, the first year for which domestic investment is available is 1976-77.

¹⁶ This includes not only the four pre-sample years (T=4) but also the first two sample years ending in 1977-78 - roughly the middle of the 1976-77 to 1979-80 period used to compute firms' average investments in the first step.

considerable amounts in accumulating technical knowledge. Thus, this paper also considers an additional grouping consisting of firms in these four "technology intensive" industries. To check whether and to what extent the results are driven by pooling across two-digit industries, separate estimates by two-digit industry are also presented.

4. Empirical Findings

Tables 2a and 2b detail estimates from the standard fixed and random effects models for both the Cobb-Douglas and the Translog production functions for various firm groupings. There are several significant features of the estimates. First, in every set of estimates the Hausman statistic rejects the random effects specification, suggesting that unobserved time-invariant firm heterogeneity does influence the input choices of firms. Second, the estimated coefficients pertaining to the conventional inputs, capital, labor, and materials lie in ranges conforming with those found by virtually every study which utilizes firm-level data to estimate production relationships while controlling for firm specific heterogeneity.¹⁷ Additionally, the estimated returns to scale in terms of these inputs are reasonable. For the Cobb-Douglas production function, returns to scale range from 0.95 to 1.02 for the various estimates. In the case of the Translog production function where output elasticities are not constant over the estimated equation but vary with individual observations, average returns to scale lie in a similar range.¹⁸ Taken together these findings suggest that the estimated relationships are fairly sensible. Finally, a focus on the fixed effects estimates reveals that the estimates of the various technology related variables are fairly similar across the Cobb-Douglas and Translog functional forms. In fact, the estimates for all firms and scientific firms are virtually identical: exactly the same set of variables are statistically significant and for these variables even the sizes of the estimated parameters are very similar.

The various sets of estimates reveal a positive and generally statistically significant impact of new capital goods on productivity. More specifically, the parameter estimates indicate that a doubling of the share of new domestic capital leads to a 1% to 0.3% increase in output. While the impact of a doubling of the share of new imported capital goods is less - it ranges from 0.3% to 0.1% - new imported capital goods are a much smaller share of total capital to begin with, typically about one-fourth of new domestic capital goods on average. In other words, the real impact of new imported and domestic capital goods on productivity is actually quite similar. By contrast,

¹⁷ See, for example, Tybout and Westbrook (1996) for a survey on the econometric evidence on the structure of production based on firm-level data from developing countries. In addition, see Basant and Fikkert (1996) and Raut (1995) for studies using firm-level data from India. While the latter estimates a value-added production function, the implied elasticities of capital based on value-added here are very similar to those of Raut (1995). Moreover, in the production functions estimated here the total impact of capital on output is spread across the capital stock variable and the variables measuring the new technology embodied in recently purchased capital goods.

¹⁸ The Translog estimates of output elasticities for capital, labor, and materials are obtained by evaluating the estimated production function at the mean values of those variables. T-statistics were obtained using the delta method. For all four industry groupings, the estimated production functions display generally good curvature properties. For example, returns to scale from the preferred fixed effects model decline with output as theory would suggest and lie within conventionally reported ranges.

investments in technical knowledge only have a positive and statistically significant impact on productivity if they are foreign in origin. Firms' own R&D never shows up with a positive and statistically significant sign in any of the fixed effects estimates although they do so in some of the random effects estimates.

4.1 Autocorrelated Errors

The coefficients from the fixed effects model (and random effects model) above are estimated assuming that the idiosyncratic component of the error term, v_{it} , is independent across firms and over time. While the assumption of independence of this error term across firms is fairly innocuous, the same cannot be said about the assumption of its independence over time. In other words, v_{it} may be autocorrelated. Autocrorrelated errors can arise for several reasons. For instance, an important independent variable that itself is autocorrelated may be omitted. Alternatively, random shocks that impact firms' output and productivity, such as a strike, may have effects that persist over time. As is well known, ignoring the autocorrelation of the error term when it is present renders estimates inefficient. It is thus important to test for autocorrelated errors and, if they are present, correct for them.

Assuming first order autocorrelated (AR1) errors, i.e., $v_{it} = \rho v_{it-1} + \eta_{it}$ where ρ is the autocorrelation coefficient and η_{it} is an i.i.d. error, it is possible to utilize the procedure of Bhargava, Franzini, and Narendranathan (1982) to test for autocorrelation in the fixed effects model.¹⁹ In every case, the Bhargava et al procedure leads to a rejection of the null of no autocorrelation making it necessary to correct for AR1 errors. This is done following the standard two step procedure whereby, first, a consistent estimate for ρ is obtained by estimating the model ignoring the autocorrelation, and second, efficient estimates are obtained for the production function using Cochrane-Orcutt transformed data (for example, $x_{it}^*=x_{it} - \rho' x_{it-1}$, where ρ' is the estimated

¹⁹ Using the Durbin-Watson statistic generalized to panel data, $D_p = ('_{i=1}'_{t=2} (v_{it} - v_{it-1})^{2/'}_{i} (v_{it}^2)^{2}$ where i runs from 1 to N and t runs from 2 to T and 1 to T in the numerator and denominator, respectively, Bhargava, Franzini, and Narendranathan (1982) develop lower (D_{pl}) and upper bounds (D_{pu}) on D_p that vary only with the number of panels and time-periods. Analogous to the standard Durbin-Watson test, the null hypothesis of no (positive) autocorrelation is rejected if the sample criterion D_p is less than D_{pl} . It is accepted if D_p is greater than D_{pu} . The test is inconclusive if $D_p < D_p < D_{pu}$.

autocorrelation coefficient).20,21

Tables 3a and 3b detail the fixed effects estimates corrected for AR1 errors for both the Cobb-Douglas and Translog production functions.²² The output elasticities of the various inputs are once again broadly similar across the two production functions. However, the greater flexibility afforded by the Translog appears to come at the expense of precision as evidenced by statistically insignificant estimates on some of the technology inputs that were statistically significant before. Additionally, the estimated Translog production function for the non-scientific firms is not as well behaved as before: estimated returns to scale in terms of the conventional inputs increase with size. Focusing then on the more sensible Cobb-Douglas estimates, it can be seen that the only technology variable that does not change much in either its estimate or statistical significance is the share of new domestic capital. By contrast, the share of new imported capital retains its statistical significance only in the case of the high technology group of firms. As for the imports of disembodied technology, these remain statistically significant in the scientific and technology intensive groupings and when all firms are pooled together.

In broad terms, the pattern that emerges is one where improvements in technology embodied in new domestic capital goods are most easily translatable into improvements in productivity while it is only firms in more technologically oriented industries that are able to utilize their imports of disembodied technology and new capital goods towards improving productivity significantly. While this pattern is a reasonable one - it is entirely plausible that India would lag behind developed countries to a greater degree in the more technologically oriented industries and/or that the transferability of technology would be greatest in these industries - the fact that the broad-based impact of new capital goods stems on account of domestically purchased capital goods is a little puzzling. However, it has to be acknowledged that new domestic capital goods could also embody

$$\rho^{\prime} \stackrel{\prime}{} \stackrel{}{\overset{(j)}{=}} \frac{ [j_{i 1} \quad j_{i 2} \quad \hat{v}_{it} \mathcal{O}_{it\&1}] / N \mathcal{O} T \& 1)]}{\frac{N}{1} \quad \frac{1}{1} \quad \frac{N}{1} \quad \frac{T}{1} \quad \hat{v}_{it}^{2}] / [N \mathcal{O} T \& 1) \& K]}$$

where a hat (^) over the v_{it} 's represents estimated values of the error term from the original model ignoring autocorrelation.

²¹ If simultaneity between output and inputs is also a concern, then the estimate of ρ so obtained will not be consistent. In practice, addressing simultaneity is very difficult in view of the lack of suitable instruments. See Appendix 2, however, for an attempt to address issues of simultaneity in the presence of autocorrelated errors using GMM-IV techniques applied to panel data.

²² Random effects models corrected for AR1 errors were also estimated. Of course, these would still be susceptible to an omitted variable bias on account of time-invariant firm effects and to simplify the exposition, are not reported.

²⁰ The estimated autocorrelation coefficient is obtained as:

superior foreign technologies. In fact, they are very likely to do so as Tables 1a-c show: Indian capital goods producers, i.e. firms in the electrical and non-electrical machinery and transportation producing industries, import a considerable amount of disembodied technology and tend to do so most intensively. To the extent that these imports translate into domestically produced capital goods of higher quality, the latter would be very likely to improve productivity in user industries. Support for this possibility is found from the production function estimates obtained by pooling firms into more disaggregated groupings.

Table 4a presents the estimates from Cobb-Douglas production functions obtained by pooling firms into 8 approximately two-digit industries.²³ Unfortunately, in two of the industries, the estimated coefficients on capital are negative suggesting strongly that while choosing more disaggregated industry groupings limits the possibility of biases from pooling over potentially heterogenous industries, the inevitably smaller size of the sub-samples increases the potential for small sample biases. More sensible estimates are instead obtained by constraining the conventional inputs, capital, labor and materials, to have the same relationship to output across more aggregated industrial groupings while allowing the impact of the technology related variables to vary across the more disaggregated two-digit industries. Table 4b presents such estimates.

It is encouraging to find that there are many similarities in the estimates pertaining to the technology inputs across Tables 4a and 4b. For example, out of a total of 32 estimates for the technology inputs (8 industries with four technology inputs each), only four have different signs across the two sets of estimates. More importantly, the main effect of constraining the estimates of K, L, and M to have the same coefficients across two-digit industry groups seems to be to lower standard errors thereby improving the statistical significance of various estimates (there are only three exceptions to this - two in chemicals and one in non-electrical machinery).

Focusing, then on the estimates of Table 4b, it can be seen that whenever the share of new capital goods is statistically significant it is positive. Additionally, while the positive and significant impact of new domestic capital goods is found in a range of industries - those deemed scientific, non-scientific, or technology intensive - they have a larger impact on productivity in the industries that may be classified as scientific and/or technology intensive, just as the estimates of Table 3a indicated. Similarly, while new imported capital goods impact productivity positively and significantly in only two industries, transportation and non-electrical machinery (and chemicals if the estimate of Table 4a are considered), both (and all three if Table 4a were considered) are industries which may be classified as scientific and/or technology intensive.

A similar pattern of relative importance in productivity improvements across industries is evident for imports of disembodied technology. These have a positive impact on productivity in the electrical and non-electrical machinery industries (and nearly so in chemicals if the estimates of Table 4a are considered), once again industries which may be classified as scientific and/or

²³ Translog production functions were also estimated for each of these eight industries. The estimated elasticities for the various inputs were again quite similar to those from the Cobb-Douglas production functions. However, the curvature properties were not very good in several cases.

technology intensive.²⁴ Moreover, Table 4b (or even Table 4a) also shows why the non-scientific industries of Table 3a show a statistically insignificant impact of disembodied imported technology on productivity: while these have a positive and significant impact in the textile industry, they have a negative and significant impact in the metal producing industries (although this latter impact is statistically insignificant in the estimates of Table 4a). Similarly, Table 4b (or even Table 4a) also shows why R&D fails to show up with a positive and statistically significant impact in any of the industrial groupings of Table 3a: the individual two-digit industries do not reveal much of a positive and significant impact of R&D stocks on productivity.

What is responsible for the broad-based lack of a positive and statistically significant impact of R&D on productivity? Investments in disembodied technology can entail a fair degree of uncertainty and risk. Thus a firm which conducts R&D in order to streamline its production process, for example, may simply fail to do so. In the best case scenario it would still be able to revert to its previously standardized process and produce as much output given its other inputs. The lack of any statistical relationship between its R&D and output would be consistent with such a scenario. In the worst case scenario the new, failed production process may not be easily reversed to the previous process and the actual production may fall short of what was previously possible given the same inputs. In this case there would appear to be a negative relationship between R&D and output.

Although R&D efforts may fail a significant number of times, it still must be explored whether there are other reasons for the lack of a connection between these and productivity. One possibility that needs to be explored is whether there are important complementarities between the various technology inputs. Thus while an Indian firm conducting R&D in isolation may end up with what in effect turns out to be a misdirected effort, it is still possible that R&D conducted in the context of imports of disembodied technology may lead to productivity gains. This possibility is examined next.

4.2 Interactions between Technology Variables

As mentioned in the introduction, one issue of long-standing interest among economists and policymakers in developing economies is whether significant investments in developing local technological capability - as acquired through formal R&D activities, for example - are required for utilizing imported technology effectively. Various researchers have argued in the affirmative and even claimed that foreign technologies are often the seed for the development of local technological capability.²⁵ Others have instead argued that no significant local capabilities are required in using imported technology effectively and that imports of technology may even dampen local efforts. While one way to examine these views is by modeling and estimating firms' choices regarding investments in technology using a factor demand framework as in Fikkert (1996) and Raut (1988),

²⁴ Note also that both of these industries form the backbone of the capital goods manufacturing industries thereby lending support to the hypothesis above that domestic capital goods may well incorporate productivity enhancing foreign technologies.

²⁵ See, for example, Katrak (1989) and Siddharthan (1992).

another is to estimate the relationship between domestic and imported technology inputs in production as in Basant and Fikkert (1996).²⁶ This section extends the analysis of the latter by introducing as additional regressors interaction terms involving the various technology inputs. If an interaction term between two variables, say R and T shows up positively (negatively), then the two share a complementary (substitute) relationship in the sense that an increase in R or T leads to an increase (decrease) in the output elasticity of T or R. Thus if a significant amount of R was needed to utilize T effectively this would tend to reveal itself in terms of a positive interaction term. If this were not the case, i.e. suppose a firm could use either R or T to obtain technical knowledge independently of the other, the interaction term would be small, zero, or even negative if diminishing returns to investing in technological knowledge were important.²⁷

Table 5a and 5b detail the estimates of Cobb-Douglas production functions estimated using the fixed effects model (with a correction for AR1 errors) for various firm groupings. As the row of estimates corresponding to the interaction term between R and T reveals in Table 5a, there is no evidence of significant complementarity between stocks of R&D and imported disembodied technology in any of the four industrial groupings. The estimates at the more disaggregated twodigit industry level, detailed in Table 5b, support this with one exception. In the electrical industry, the interaction term between R and T is positive and statistically significant, consistent with the notion that own R&D allows firms in this industry to utilize their imports of disembodied technology more effectively. In addition, the estimates of Tables 5 a and b also indicate only limited complementarity between R&D stocks and imported embodied technology: of all the estimates there is only one statistically significant interaction term between R and RI^M/K (for food processing in Table 5b). On the basis of these estimates, it seems fair to say that while the imports of technology may require some in-house efforts to adapt it to local conditions, this is not as pervasive a factor in driving in-house R&D as some of the literature has claimed.

Interestingly, there is much more evidence of a complementary relationship between the shares of new imported and domestic capital goods in production: whenever the interaction terms between these shares are statistically significant in the estimates of Tables 5a and 5b, they are always positive. Apparently, new investment plans of firms require investments in both domestic

²⁶ While Fikkert (1996) finds evidence that any factor which raises imports of disembodied technology leads to a reduction of R&D, Raut (1988) finds the opposite. The two papers utilize similar data sets covering virtually the same time period (1975/76 to 1980/81). Therefore, it seems likely that an interplay of differences in methodology and variable construction are responsible. For example, although both papers use a simultaneous equation framework, Fikkert (1996) uses a maximum likelihood procedure to take into account the censoring of his two dependent variables, R&D and imported disembodied technology, at zero. While Raut (1988) does not take into account the censoring problem, he breaks down firms' expenditures on disembodied technology into five categories distinguishing, for example, between royalties and technical fees paid to foreign sources. Fikkert (1996), on the other hand, combines the payments for royalties and technical fees paid to foreign sources into one composite "technology purchase" variable.

²⁷ Suppose that either R or T could be used independently of the other for the purpose of generating useful technical knowledge. If there are diminishing returns in the generation of technical knowledge, then the larger is a firm's R or T, the smaller will be the output elasticity of R&D or imported disembodied technology. In terms of the production function estimates, this feature would show up as a negatively signed interaction term.

and imported capital goods to be most effective. Restrictions on imports of capital goods would then not only have a direct, adverse impact on productivity - as indicated by the positive and statistically significant estimate for the share of imported capital in the scientific group of industries, for example - they would also have an indirect, adverse impact by making investments in domestic capital goods less productive. This finding resonates well with the assumption made in numerous endogenous growth models that domestic and imported inputs are complements in production (Romer 1994; Lee 1995; Baldwin and Seghezza 1996). Indeed, such an assumption is crucial in these models because it provides a link whereby barriers to trade lead to inefficiencies in the accumulation of capital and ultimately tie an economy down to a lower equilibrium growth rate.

4.3 Comparison with Other Studies

There are very few studies that have estimated the impact of investments in technology on productivity using firm-level data for LDCs. In fact, those that have, have used data for India - the one LDC for which panel data on the relevant variables exists (Basant and Fikkert 1996, Raut 1995, and Ferrantino 1992). However, all three of these studies have examined only the impact of various types of disembodied technology on productivity.

Ferrantino uses panel data for 280 firms belonging to six 2 digit industries and spanning the 1975-81 period. He examines the impact of technology in the six industries by estimating an indirect cost function where technology purchase intensities enter as parameter shifters. The specification is quite general in that the technical change implied by the technology purchase intensities can lead to Hicks neutral technical change as well as affect the substitutability of capital and labor. When Ferrantino aggregates all the various components of technology into one consolidated technology share variable, he finds technology intensity to have a positive and statistically significant impact on productivity in only two industries: chemicals and light industries. However, for textiles and machinery, technology intensity has a statistically significant negative impact on productivity. By contrast, recall that the two-digit industry level estimates here indicate technology inputs to have a positive impact on productivity in the textiles industry and the two machinery producing industries (electricals and non-electricals). Moreover, the light industry estimates here yield a negative and significant impact of R&D stocks on productivity.

While it is difficult to be sure of what is driving these differences across the two papers, methodological differences could be responsible.²⁸ For example, the cost function used by Ferrantino assumes that both capital and labor are at their long-run equilibrium values. As pointed out in Fikkert and Hasan (1998), India's licensing regime imposed serious constraints on firms' choice of inputs, especially capital. Thus the appropriate cost function should be a short-run cost function which recognizes the quasi-fixity of capital. Indeed, estimating a long-run cost function

²⁸ To some extent, it is possible that the differences in the composition of the light industries may be responsible for the different estimates across the two papers for this industry. While Ferrantino includes in his light industries the producers of paper, non-metallic minerals, wood products, and leather and fur products, there are no producers of the last two product categories in the sample firms here. However, as pointed out in the text differences in methodology are playing a definite role.

in a situation where some factor is quasi-fixed induces bias in the parameters of the production technology in an indeterminate direction. Perhaps more importantly, Ferrantino does not include time effects, nor does he correct for firm-specific heterogeneity by using either a fixed effects or random effects procedure. As mentioned before, ignoring firm effects will lead to biased parameter estimates if input choices are correlated with these. Indeed, a check using data from the light industry revealed that production functions estimated without controls for time effects and firm-heterogeneity (and without corrections for autocorrelated errors) yielded a positive and statistically significant sign on the R&D stocks, consistent with the results of Ferrantino.²⁹

Basant and Fikkert (1996) and Raut (1995) estimate augmented Cobb-Douglas production functions as does this paper. The results pertaining to the impact of stocks of R&D and disembodied technology import are consistent with those of Basant and Fikkert - high returns to T and low returns to R - even when the possibility of autocorrelated errors and simultaneity bias between output and inputs are taken into account as done here (see Appendix 2 and Table A1). Basant and Fikkert also find no support for the notion that R&D and imports of disembodied technology are complements in production. While this is generally found to be the case in this paper also, the analysis at the more disaggregated two-digit industry reveals that complementarities are important for electricals industry.

Unfortunately, it is not possible to make a direct comparison of the results of this paper with Raut's because he does not treat R&D and imports of technology as distinct regressors in his production function. Instead, he pools together firms' expenditures on in-house R&D and import of disembodied technology calling it R&D. He also includes a measure of the aggregate R&D conducted by other firms in order to capture the role of R&D spillovers in each of his regressions. While Raut's disembodied technology variable (R&D plus imports of disembodied technology) is almost never statistically significant even at the 10 percent level, he does get statistically significant coefficients on the spillover variable in several of his estimates. Raut conjectures that the insignificant effect of firms' stock of disembodied technology may be due to measurement error caused by the double-counting of labor and capital used to perform R&D in the variables L and K and/or the non-reporting of R&D by some firms. He notes that the latter problem may not be there in the spillover variable since this is based on information from the Department of Science and Technology which is usually considered a reliable source of R&D data.

While the issue of double counting cannot be addressed with the available data, it does not seem likely that the low estimate of disembodied technology stocks stems from utilization of R&D numbers reported by the firms in their annual reports rather than the numbers they report to the Department of Science and Technology. As explained in Section 3, the procedure used in this paper was to cross-check the R&D numbers listed in the annual reports with those reported in the <u>Compendium of In-House R&D Centers</u> published by the Department of Science and Technology. If a firm had R&D expenditures reported in the Compendium but not in its annual report, the former

²⁹ When Ferrantino separates the various categories of technology into five categories based on expenses on R&D, royalty payments to foreigners, technical fees to foreigners, royalty payments to domestic firms, and technical fees to domestic firms, the <u>share</u> of R&D impacts productivity positively and significantly.

numbers were used. As the production function estimates in this paper reveal, even a usage of the Department of Science and Technology R&D numbers, usually considered more reliable, does not raise the impact of R&D stocks on productivity.³⁰

In addition to the fact that R&D is a difficult and risky venture which may not pay-off, the low estimates of Indian manufacturing firms' R&D, found across all of these papers, may also reflect the possibility that the Indian government's very policies to encourage private sector R&D have led firms to over-report their R&D activities. In a survey of India's R&D policy and experience, Desai (1993) argues that firms' R&D efforts were not entirely geared towards generating useful knowledge. Rather a part of the reported R&D expenditures were designed to enable firms to take advantage of the incentives - income tax breaks and favorable treatment in the granting of licenses to import technology - introduced by the Indian government for firms with R&D units certified by the Department of Science and Technology. By either interpretation of the results, the efforts of the Department of Science and Technology to raise productivity generating R&D by the private sector seem to have been ineffective.

5. Conclusion

This paper has examined the impact of firms' R&D activities, their imports of disembodied technology, and their investments in imported and domestic capital goods on productivity using a sample of Indian manufacturing firms. The estimates from production functions augmented with these technology inputs support the hypothesis that the better access to the international knowledge base that closer international integration would afford represents important opportunities for developing countries to generate productivity growth. In particular, the estimates indicate that the imports of technology, especially those of disembodied technology, make statistically significant and direct contributions to the technical efficiency of firms, especially in industries where technological opportunities and technology investments are highly prevalent - industries such as chemicals and pharmaceuticals, electrical and non-electrical machinery, and transportation equipment.

The estimates also indicate that new domestic capital goods impact productivity positively and, in fact, do so for a broader range of manufacturing industries. However, the productivity enhancing impact of new domestic capital goods itself seems to stem from imported technology: firms belonging to the domestic capital goods industry tend to be the largest and most intensive users of imported disembodied technology. Moreover, there is evidence of a complementary relationship between imported and domestic capital goods in production in a number of the specifications estimated in this paper suggesting that restrictions on firms' import of capital goods, in addition to restricting access to embodied foreign technology, may have had another adverse effect on firms' productivity: lower technical efficiency of domestic capital goods as a result of lack of complementary inputs.

³⁰ R&D stocks based on only the expenditures reported in the firms' annual reports were not associated with higher productivity either.

In contrast to the other technology inputs, the impact of own R&D is found to be quite low. Although this may be a result of firm heterogeneity in ability to perform R&D - the random effects estimates in this paper yield larger and sometimes statistically significant impacts on productivity they could also result from measurement problems. In particular, the incentive schemes introduced by the Indian government to encourage R&D may have led firms to seriously over report their R&D activities. To the extent that this happened, it points to a serious flaw in the ability to implement such government efforts aimed at encouraging firms to perform R&D. Indeed, the numbers presented in this paper suggest that India's liberalization of its trade and industrial policies, begun in earnest in 1991 but still ongoing, may lead to a decline in firms' R&D efforts as firms' increase their expenditures on imported technology in the more liberalized regime. The decline in R&D may be arrested to the extent that some R&D is needed to utilize imported technologies more effectively. However, the limited complementarity found in this paper between R&D and imported technologies suggests that this is not likely to be as important a countervailing force.

In other words, proponents of India's ongoing liberalization based reform efforts should be cautious when they argue that the pro-competitive forces of liberalization will lead local firms to conduct more R&D. Reductions in R&D in the more liberalized environment may well give ammunition to opponents of liberalization, especially those who tend to look at any substitution of local efforts at technology generation with imports with great suspicion (and called "technology nationalists" by Desai, 1993). A more constructive approach, not only from the point of firms' productivity but also with regard to the generation of local technological capability, may be to accept that foreign technological capability must focus on tools other than import substitution. As Bell and Pavitt (1993) point out, local technological capability has for too long been viewed solely through the lens of trade policy in development debates.³¹ Over time India has accumulated a vast body of scientific and technical personnel in government owned scientific departments and institutions of higher learning. Efforts at aligning them more closely with private sector requirements and making them less dependent on government finance and management may well be challenging, but they could lead to large pay-offs.

³¹ Thus those on the right often argued that an open trade policy would encourage the development of local technological capability while those on the left argue that local technological capability would only be harnessed in closed regimes (Bell and Pavitt,1993).

<u>Appendix 1</u> <u>Construction of Capital Stocks</u>

The procedure used in Fikkert and Hasan (1998) was followed to construct a firm specific capital stock variable, K_t , that is net of depreciation and expressed in constant 1985-86 rupees. The first step was to compute the average age of the capital stock in the first year for which a firm's data is available. This was done by using values of accumulated depreciation (AD) and total gross fixed assets (TGFA), both of which are in the data set and by making the assumption that full depreciation of a firm's capital stock takes 16 years for accounting purposes. In this case, the average age (AA_i) of a firm's capital stock is:

$$AA_i \stackrel{'}{=} \frac{AD_i}{TGFA_i} @6 \tag{A1.1}$$

If it is further assumed that all of the capital stock in the first year has been purchased AA_i years ago the capital stock in 1985-86 rupee value becomes:

$$K_{i0} \stackrel{\prime}{=} \frac{TGFA_{i0}}{P_{K, 0\&AA_i}} \mathfrak{P} \& \delta]^{AA_i}$$
(A1.2)

where δ denotes the rate of depreciation of the capital stock, t=0 represents the first year for which a firm's data is available, and P_{Kt} is the value of the investment deflator in year t. In all future time periods (t > 0), the firm's capital stock evolves as:

$$K_{it} \stackrel{\prime}{} I_t \,\,\% \, K_{it\&1} \, \Phi 1 \,\& \delta] \tag{A1.3}$$

where I_t denotes real investment and has been calculated as the difference in TGFA between years, deflated by P_{Kt} . The rate of depreciation was chosen to be 6 percent per annum as in Fikkert and Hasan.

Another way to compute firm- and time-specific net capital stocks is to simply deflate each firm's total gross fixed assets by the investment deflator:

$$K_{it} - \frac{TGFA_{it}}{P_{K,t}}.$$
 (A1.4)

Whereas the procedure outlined in equations (A1.1) - (A1.3) makes the implicit assumption that there is no revaluation of fixed assets, the capital stocks of equation (A1.4) are based on the assumption that revaluations are made every year on the basis of replacement cost and are undertaken so that the fixed assets recorded in the balance sheet may reflect their intrinsic value.

The two capital stocks, i.e., the one based on the perpetual inventory method and that based on equation (A1.4), move closely together and using either of the capital stock measures in the empirical analysis does not lead to qualitatively different results (Hasan 1997).

<u>Appendix 2</u> <u>An Exploration of Endogeneity Issues Using GMM-IV Estimation</u>

The various estimates presented in the paper are obtained under the assumption that the regressors - the various production inputs - are uncorrelated with the error term. If changes in output and inputs are correlated - as they would be if they were simultaneously determined - then the estimates would be subject to a simultaneity bias. Such a bias could occur on account of either demand shocks or liquidity shocks, for example.

In principle, the way to tackle the simultaneity problem is through the usage of an instrumental variable estimator. In practice, however, finding an instrument which is correlated with the independent variable(s) being instrumented but not the error term can be very difficult, particularly in the context of micro-level data such as used here. Panel data are useful in such cases because they present a potentially useful instrument: lagged values of the regressors themselves (see, for example, Griliches and Hausman 1986, Mairesse and Hall 1996, Arellano and Bond, 1991). This appendix presents estimates which control for simultaneity bias and omitted variable bias due to time-invariant firm effects. In particular, time-invariant firm effects are dealt with by first differencing the data and simultaneity is dealt with by using lagged values of right hand side variables as instruments. Estimation is via the generalized method of moments (GMM) procedure of Arellano and Bond (1991) which provides a gain in efficiency as compared to the first differenced instrumental variables method suggested by Anderson and Hsiao (1981) for the case of dynamic fixed-effects models.

As Arellano and Bond point out, estimators that assume i.i.d. errors and utilize lagged values of the potentially endogenous right hand side variables as instruments would lose their consistency if the errors were actually autocorrelated. Because the analysis of Section 4 revealed the presence of autocorrelation, it is necessary to first transform the regression equation so as to eliminate the autocorrelation. Thus if the original model is:

$$y_{it} \stackrel{\prime}{} \beta \mathfrak{R}_{it} \, \% \, u_i \, \% \, v_{it}, \tag{A2.1}$$

where v_{it} is an AR1 process ($v_{it} = \rho v_{it-1} + \eta_{it}$; ρ is the autocorrelation coefficient and η_{it} is an i.i.d. error), then it is possible to re-express the equation in dynamic form as:

$$y_{it} \circ \mathfrak{g}_{it\&1} \otimes \mathfrak{g}_{it} \otimes \mathfrak{g}_{it\&1} \otimes \mathfrak{g}_{i$$

where η_{it} is i.i.d. as before and non-linear common factor restrictions are imposed on the values of β and ρ . The term involving u_i may be removed by differencing the data. Moreover, as η_{it} is i.i.d., any correlation between the differenced right hand side variables, Δy_{it-1} , Δx_{it} , and Δx_{it-1} , and the differenced error term, $\Delta \eta_{it}$, may now be dealt with by choosing lagged levels of the differenced right hand side variables dated t-2 or earlier as instruments.

For estimation purposes, each of the right hand side variables is assumed to be potentially endogenous and is instrumented for using lagged values of all the right hand side variables except for capital. In the case of capital, it is assumed that in addition to simultaneity, there may be endogeneity on account of mis-measurement of capital.^{32, 33} If true this would invalidate the usage of lagged values of capital as an instrument because by virtue of its construction, mis-measurement in capital at any given point of time would be carried over to the future. As argued by Westbrook and Tybout (1993) and Tybout (1992), items such as land and buildings rather than purchases of plant and equipment usually account for a large part of the error in measuring capital thereby rendering investments in plant and equipment as a possible instrument for capital. However, because simultaneity between current investments in plant and equipment and output is of concern, this paper uses lagged values of these investments as instruments. Finally, lagged values of real wages are also included as instruments as their movements are likely to have an impact on firms' decisions regarding their capital stocks.

Table A1 presents the GMM-IV estimates using the method of Arellano and Bond.³⁴ Because the procedure is very intensive in data, estimates are obtained using the more aggregated industrial classification, i.e., for all firms and the scientific, non-scientific, and technology-intensive industries. The table also presents results from two tests of instrument validity due to Arellano and Bond. These are a test for the absence of *second* order serial correlation, since by definition, first differencing will introduce first order serial correlation in the differenced error term: $\Delta \eta_{it} = \eta_{it} - \eta_{it-1}$) and a Sargan test of over-identifying restrictions. The test statistic for second order serial correlation - based on the GMM residuals from the first-difference equation - is distributed as a standard normal with zero mean and unit variance. As the table reveals, the test statistic leads to an acceptance of the null of no second order serial correlation in all four sets of estimates. Similarly, the Sargan test of over-identifying restrictions, which is distributed as chi-squared under the null that the instruments are valid, also leads to a validation of the instruments.

The actual estimates are qualitatively similar to those of the fixed effects model corrected for autocorrelation (Table 3a of Section 4). The imports of disembodied technology are positive and statistically significant for all firms and the scientific and technology intensive industries, while the

³² To see this, suppose that $K_{it} = K_{it}^* + w_{it}$, where K_{it}^* is the true but unobserved capital stock and w_{it} is i.i.d. measurement error. Suppose that we estimate equation (A2.2) using ΔK_{it} rather than ΔK_{it}^* , i.e. $\Delta K_{1it} = K_{it} - K_{it-1}$. Then under the assumption that w_{it} is uncorrelated with η_{it} , the actual error term of the first-difference equation is $\Delta(\eta_{it} - \beta_K \circledast_{it})$ which is correlated with ΔK_{it} since $\Delta K_{it} = \Delta(K_{it}^* + w_{it})$. The result will be a downward bias in the coefficient on capital.

³³ The same problem may afflict the other two stock variables included in the production function, R and T. However, obtaining suitable firm and time specific instruments for these technology variables with the available data was not possible. In any case, as is shown below the tests of instrument validity did not rule out the usage of lagged R and T as instruments.

³⁴ The estimates are based on Arellano and Bond's one-step estimates. Consistent with Arellano and Bond's warning that while their two-step estimates are asymptotically more efficient they are susceptible to spuriously low standard errors in small samples, the two-step estimates obtained from the application here had implausibly low standard errors.

share of new domestic capital goods are positive and statistically significant for all firms and nonscientific group. The share of new imported capital goods fails to be statistically significant in any of the estimates, however. Other than that, the main difference between the GMM-IV estimates and those of the fixed effects model with correction for AR1 errors is in terms of the estimates on capital. While the estimates of capital in the scientific and technology intensive industries are statistically significant (and substantially larger than previous estimates), they are statistically insignificant for all firms and the non-scientific industry. Although this is an unfortunate feature of the GMM-IV estimates, the relative robustness of the findings for the technology related variables across very different methodologies is encouraging.

REFERENCES

Ahluwalia, I. J. (1994). "The Role of Trade Policy in Indian Industrialization" in G.K. Helleiner edited, <u>Trade Policy and Industrialization in Turbulent Times</u>. London; New York: Routledge.

Anderson, T.W. and C. Hsiao (1981). "Estimation of Dynamic Models with Error Components," Journal of the American Statistical Association, vol 76, pp. 598-606.

Arellano, M. and S. Bond (1991). "Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations," <u>Review of Economic Studies</u>, vol 58, pp.277-97.

<u>Annual Survey of Industries: Summary Results for the Factory Sector.</u> Central Statistical Organization, Ministry of Planning, Government of India, New Delhi, published annually.

Arora, A. (1996). "Contracting for Tacit Knowledge: The Provision of Technical Services in Technology Licensing Contracts," Journal of Development Economics, vol. 50, pp. 233-256.

Bahk, B. H. and M. Gort (1993). "Decomposing Learning by Doing in New Plants," Journal of Political Economy, vol. 101, no. 4, pp.561-583.

Baily, M., C. Hulten, and D. Campbell (1992). "Productivity Dynamics in Manufacturing Plants."<u>Brookings Papers on Economic Activity: Microeconomics 1992</u>. The Brookings Institution, Washington D.C.

Baldwin, R. and E. Seghezza (1996). "Testing for Trade-Induced Investment Led Growth," <u>NBER</u> <u>Working Paper #5416</u>.

Basant, R. and B. Fikkert (1996). "The Effects of R&D, Foreign Technology Purchase, and International and Domestic Spillovers on Productivity in Indian Firms," <u>The Review of Economics</u> and <u>Statistics</u>, vol. 78, no. 2, pp. 187-199.

Bell, M. and K. Pavitt (1993). "Accumulating Technological Capability in Developing Countries," <u>World Bank Economic Review</u>, The World Bank, Washington DC.

Berndt, E. and C. J. Morrison (1995). "High-tech Capital Formation and Economic Performance in U.S. Manufacturing Industries: An Exploratory Analysis," <u>Journal of Econometrics</u>, vol. 65, pp. 9-43.

Bhargava, A., L. Franzini and W. Narendranathan (1982). "Serial Correlation and the Fixed Effects Model," <u>Review of Economic Studies</u>, October.

Bregman, A., M. Fuss, and H. Regev (1991). "High-Tech and Productivity: Evidence from Israeli Industrial Firms," <u>European Economic Review</u>, 35, pp. 1199-1221.

Chandhok, H. L. and the Policy Group (1990). India Database: The Economy, Volumes I-II.

Clerides, S., S. Lach, and J. Tybout (1998). "Is 'Learning-by-Exporting' Important? Micro-Dynamic Evidence from Colombia, Mexico, and Morocco." <u>Quarterly Journal of Economics</u>.

Coe, D. and E. Helpman (1993). "International R&D Spillovers," <u>European Economic Review</u>, vol. 39, pp. 859-887.

<u>Compendium on In-House R&D Centers</u>, Department of Science and Technology of India. New Delhi: Department of Science and Technology, published annually.

Cueno, P. and J. Mairesse (1984). "Productivity and R&D at the Firm Level in French Manufacturing." In Z. Griliches edited, <u>R&D</u>, <u>Patents</u>, and <u>Productivity</u>. Chicago: The University of Chicago Press.

De Long, J.B. and L.H. Summers (1993) "How Strongly Do Developing Economies Benefit from Equipment Investment?" Journal of Monetary Economics, 32, pp. 395-415.

De Long, J.B. and L.H. Summers (1991) "Equipment Investment and Economic Growth," <u>Quarterly</u> <u>Journal Economics</u>, 106, pp. 445-502.

Desai, A. (1993). "The R&D System of India," unpublished paper.

Desai, A. (1988). Technology Absorption in Indian Industry. Wiley Eastern: New Delhi.

Directory of Foreign Collaborations in India, India Investment Center, Government of India.

Ferrantino, M. (1992). "Technology Expenditures, Factor Intensity, and Efficiency in Indian Manufacturing." <u>The Review of Economics and Statistics</u>, vol. 74, pp. 689-700.

Fikkert, B. and R. Hasan (1998). "Returns to Scale in a Highly Regulated Economy: Evidence from Indian Firms." Journal of Development Economics, vol. 56, pp. 51-79.

Fikkert, B. (1996). "An Open or Closed Technology Policy?: The Impacts of International Knowledge Diffusion on Self-Reliance and Productivity in Less Developed Countries," working paper at the University of Maryland, revised March 1996.

Fikkert, B. (1994). <u>An Open or Closed Technology Policy?: India's Regulation of</u> <u>Technology Licenses, Foreign Direct Investment, and Intellectual Property</u>, unpublished Ph.D. Dissertation at Yale University.

Griliches, Z. and J. Hausman (1986). "Errors in Variables in Panel Data." Journal of Econometrics, 31, pp. 93-118.

Griliches, Z. and J. Mairesse (1984). "Productivity and R&D at the Firm Level," in Z. Griliches edited, <u>R&D</u>, <u>Patents and Productivity</u>, pp. 339-374, University of Chicago Press.

Grossman, G. and E. Helpman (1991). <u>Innovation and Growth in the Global Economy</u>, Cambridge, MA: MIT Press.

<u>Handbook of Industrial Statistics</u> (1987). Office of the Economic Advisor, Department of Industrial Development, Ministry of Industry, Government of India, New Delhi, published annually.

Hasan, R. (1997). <u>Productivity Growth and Technological Progress in a Reforming Economy:</u> <u>Evidence from India</u>. Unpublished Ph.D. dissertation at the University of Maryland, College Park.

Hulten, C. R. (1992). "Growth Accounting When Technical Change Is Embodied in Capital," <u>American Economic Review</u>, vol. 82, no. 4, pp. 964-980.

Kapur, S.L. (1983). "Import of Technology by India: Policy, Procedures, and Problems." Working paper at Center for Monitoring the Indian Economy, Bombay, India.

Katrak, H. (1997). "Developing Countries' Imports of Technology, In-House Technological Capabilities and Efforts: An Analysis of the Indian Experience." Journal of Development Economics, vol. 53, pp. 67-83.

Katrak, H. (1989). "Imported Technologies and R&D in a Newly-Industrializing Country: The Experience of Indian Enterprises," Journal of Development Economics, 31, pp. 123-139.

Kumar, N. (1990). "Cost of Imported and Local Technologies: Implications for Technology Policy," <u>Economic and Political Weekly</u>, January 13, pp. 103-106.

Lall, S. (1987). Learning to Industrialize. London: The Macmillan Press.

Lee, J. W. (1995). "Capital Goods Imports and Long-Run Growth," Journal of Development Economics, vol. 48, pp. 91-110.

Mairesse, J. and B. Hall (1996). "Estimating the Productivity of R&D: An Exploration of GMM Methods Using Data on French and U.S. Manufacturing Firms," <u>NBER Working Paper #5501</u>.

Pandit, B.L. and N.S. Siddharthan (1995). "Technological Acquisition and Investment: Lessons form Recent Indian Experience," Working Paper.

Raut, L. (1995). "R&D Spillover and Productivity Growth: Evidence from Indian Private Firms," Journal of Development Economics, vol. 48, pp.1-23.

Raut, L. (1988). "R&D Behavior of Indian Firms: A Stochastic Control Model," <u>Indian Economic</u> <u>Review</u>, vol 23, no. 2, pp. 207-229. <u>Research and Development Statistics</u>, Department of Science and Technology, Government of India, New Delhi, published annually.

Rodrik, D. (1992). "Closing the Productivity Gap: Does Trade Liberalization Really Help?" in G. Helleiner, ed. <u>Trade Policy</u>, <u>Industrialization</u>, and <u>Development</u>. Oxford: Clarendon Press.

Romer, P. (1994). "New Goods, Old Theory, and the Welfare Costs of Trade Restrictions," Journal of Development Economics, vol. 43, pp. 5-38.

Siddharthan, N. S. (1992). "Transaction Costs, Technology Transfer, and In-House R&D. A Study of the Indian Private Corporate Sector." Journal of Economic Behavior and Organization, 18, pp. 265-271.

Scott-Kemmis, D. and M. Bell (1988). "Technology Import Policy: Have the Questions Changed?" in A. Desai edited, <u>Technology Absorption in Indian Industry</u>. Wiley Eastern: New Delhi.

Statistical Abstract of India, Government of India, New Delhi, published annually.

Tybout, J. and Westbrook, D. (1996). "Scale Economies as a Source of Efficiency Gains." in <u>Industrial Evolution in Developing Countries: Micro Patterns of Turnover, Productivity and Market</u> <u>Structure</u>, edited by M. Roberts and J. Tybout, chap. 5, Oxford University Press.

Tybout, J. (1992). "Making Noisy Data Sing: Estimating Production Technologies in Developing Countries." Journal of Econometrics 53, pp. 25-44.

Westbrook, D. and Tybout, J. (1993). "Estimating Returns to Scale with Large, Imperfect Panels: An Application to Chilean Manufacturing Industries." <u>The World Bank Economic Review</u>, vol. 7, No. 1, pp. 85-112.

Young, A. (1991). "Learning by Doing and The Dynamic Effects of International Trade." <u>Quarterly</u> <u>Journal of Economics</u>, 106, pp. 369-405.

Table 1a
Basic statistics by two-digit industry

Industry	Firms	Output (Q)	Capital (K)	Labor (L)	Materials (M)
Food	15	210,136	127,304	18,833	119,331
Textiles	48	521,903	387,101	84,599	259,748
Light	34	744,268	639,233	70,377	274,616
Metals	22	438,375	233,984	36,406	235,743
Chemicals	80	493,960	408,475	52,258	280,809
Transportation	25	1,165,106	676,990	147,334	705,510
Electricals	33	548,033	209,641	76,596	360,310
Non-Electricals	29	419,531	198,367	54,198	249,614

Note: Numbers are mean values over firm means for Q, K, L, and M. All numbers are in thousands of constant rupees.

Industry	R&D Capital	Imported	Recent Investments	Recent Investments
	(R)	Disembodied	in Imported Capital	in Domestic Capital
		Technology	(RI^{M})	(RI^{D})
		Capital (T)		
Food	0	0	1	20,059
Textiles	0	0	3,845	58,802
Light	1,212	671	10,301	99,956
Metals	54	1,645	2,491	26,556
Chemicals	4,197	201	3,108	47,855
Transportation	2,346	5,074	25,952	77,950
Electricals	2,267	5,821	7,854	28,456
Non-Electricals	674	5,383	5,093	41,197

Note: Numbers are median values over firm means for R, T, RI^M, and RI^D. All numbers are in thousands of constant rupees.

Industry	R>0	T>0	RI ^M >0	RI ^D >0
Food	0.40	0.13	0.53	1.00
Textiles	0.35	0.44	0.98	1.00
Light	0.74	0.68	0.97	1.00
Metals	0.50	0.86	0.86	1.00
Chemicals	0.84	0.63	0.99	1.00
Transportation	0.84	0.88	1.00	1.00
Electricals	0.88	0.88	0.94	1.00
Non-Electricals	0.59	0.93	1.00	1.00

Table 1b Fraction of firms with positive technology inputs

Table 1c Intensities of technology inputs

Industry	R/Q	T/Q	RI ^M /K	RI ^D /K
Food	0.000	0.000	0.000	0.180
Textiles	0.000	0.000	0.023	0.301
Light	0.003	0.001	0.032	0.257
Metals	0.000	0.004	0.022	0.227
Chemicals	0.015	0.001	0.018	0.256
Transportation	0.007	0.012	0.082	0.280
Electricals	0.007	0.012	0.049	0.229
Non-Electricals	0.001	0.017	0.039	0.324

Note: Median value over firm means

Variables	All I (2)	Firms 86)	Scientif	ic Firms 13)	Non-Scien	tific Firms 73)	Technolog Firms	y-Intensive (167)
	Fixed	Random	Fixed	Random	Fixed	Random	Fixed	Random
	Effects	Effects	Effects	Effects	Effects	Effects	Effects	Effects
ln(R)	0.000 ^a	0.001	0.001	0.001	0.000 ^a	0.001**	0.001	0.001
ln(T)	(-0.37)	(0.83)	(0.69)	(1.06)	(-0.26)	(1.64)	(1.10)	(1.37)
	0.005*	0.004*	0.010*	0.007*	0.002**	0.002**	0.009*	0.007*
$\ln(RI^{M}/K)$	(5.96)	(5.32)	(6.56)	(5.20)	(1.79)	(1.77)	(7.23)	(6.62)
	0.002*	0.001	0.001	0.001	0.003*	0.002*	0.003*	0.003*
$\ln(RI^{D}/K)$	(2.06)	(1.60)	(0.54)	(0.42)	(3.31)	(2.93)	(2.09)	(2.00)
	0.006*	0.006*	0.014*	0.015*	0.006*	0.006*	0.014*	0.016*
	(3.35)	(3.40)	(2.21)	(2.49)	(3.66)	(3.73)	(2.64)	(3.06)
ln(K)	0.05*	0.08*	0.08*	0.08*	0.05*	0.09*	0.04*	0.06*
	(5.01)	(10.04)	(4.29)	(5.72)	(4.97)	(9.67)	(2.47)	(5.01)
ln(L)	0.29*	0.27*	0.25*	0.21*	0.36*	0.32*	0.268	0.24*
	(19.87)	(22.48)	(9.80)	(11.02)	(22.00)	(24.04)	(12.88)	(15.15)
ln(M)	0.62*	0.61*	0.69*	0.68*	0.55*	0.54*	0.67*	0.65*
Adjusted R ²	(79.72)	(83.91)	(57.16)	(60.71)	(58.21)	(62.07)	(64.5 <i>3</i>)	(68.58)
	0.98	0.91	0.98	0.93	0.99	0.92	0.98	0.93
Hausman Statistic		144.40		34.06		169.40		59.61

Table 2a Cobb-Douglas production functions (Time dummies included in all regressions)

Note: T-statistics are in parenthesis. All Hausman Statistics are significant at the 1 percent level. Constant term not reported for random effects model. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Variables	All I (2)	Firms 86)	Scientif	ic Firms 13)	Non-Scien	tific Firms 73)	Technolog Firms	y-Intensive (167)
	Fixed	Random	Fixed	Random	Fixed	Random	Fixed	Random
	Effects	Effects	Effects	Effects	Effects	Effects	Effects	Effects
ln(R)	-0.001	0.001	0.002	0.003*	-0.001	0.001	0.001	0.001
	(-0.82)	(0.80)	(1.50)	(2.04)	(-1.37)	(1.04)	(1.32)	(1.57)
ln(T)	0.005*	0.004*	0.008*	0.006*	0.001	0.000^{a}	0.008*	0.006*
	(5.22)	(4.46)	(5.37)	(4.48)	(0.64)	(0.52)	(6.59)	(5.89)
$\ln(RI^{M}/K)$	0.002*	0.001*	0.002	0.002	0.002*	0.001**	0.003*	0.003*
	(2.32)	(2.00)	(1.05)	(1.20)	(2.34)	(1.83)	(2.55)	(2.75)
$\ln(RI^D/K)$	0.004*	0.004*	0.012*	0.013*	0.003**	0.003*	0.013*	0.014*
	(2.45)	(2.53)	(1.99)	(2.38)	(1.67)	(1.97)	(2.47)	(2.92)
Aln(K)/Aln(Q)	0.07*	0.10*	0.09*	0.09*	0.06*	0.10*	0.06*	0.08*
	(7.14)	(12.38)	(5.25)	(6.74)	(5.83)	(11.07)	(4.61)	(7.28)
Aln(L)/Aln(Q)	0.25*	0.25*	0.19*	0.19*	0.33*	0.31*	0.20*	0.21*
	(17.05)	(21.24)	(7.32)	(9.85)	(20.71)	(24.22)	(9.49)	(13.24)
Aln(M)/Aln(Q)	0.61*	0.59*	0.68*	0.66*	0.58*	0.56*	0.65*	0.63*
	(65.65)	(70.64)	(38.93)	(43.46)	(58.78)	(63.77)	(49.57)	(54.94)
Adjusted R ²	0.98	0.91	0.98	0.93	0.99	0.92	0.98	0.94
Hausman Statistic		239.72		40.40		261.47		74.25

Table 2b Translog production functions (Time dummies included in all regressions)

Notes: T-statistics are in parenthesis. Output elasticities have been computed at mean values and the associated t-statistics have been computed using the delta method. All Hausman Statistics are significant at the 1 percent level. Constant term not reported for random effects model. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Table 3a Cobb-Douglas production function Fixed effects model with AR1 correction (Time dummies included in all regressions)

Variables	All Firms	Scientific Firms	Non-Scientific	Technology-
	(286)	(113)	Firms (173)	Intensive Firms
				(167)
ln(R)	-0.001	0.000^{a}	-0.001	0.000^{a}
	(-1.15)	(0.20)	(-1.20)	(-0.39)
ln(T)	0.002**	0.003**	0.001	0.003*
	(1.91)	(1.86)	(0.66)	(2.58)
$\ln(RI^{M}/K)$	0.001	0.001	0.001	0.003**
	(0.88)	(0.52)	(1.22)	(1.85)
$\ln(RI^D/K)$	0.006*	0.012*	0.006*	0.013*
	(3.42)	(2.15)	(3.43)	(2.46)
ln(K)	0.03*	0.05*	0.03*	0.03*
	(2.49)	(2.73)	(2.66)	(1.98)
ln(L)	0.35*	0.31*	0.40*	0.32*
	(23.40)	(12.78)	(21.98)	(16.06)
ln(M)	0.61*	0.67*	0.53*	0.65*
	(87.89)	(72.83)	(52.10)	(79.31)
Adjusted R ²	0.97	0.97	0.97	0.97

Note: T-statistics are in parenthesis. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Table 3b Translog production function Fixed effects model with AR1 correction (Time dummies included in all regressions)

Variables	All Firms	Scientific Firms	Non-Scientific	Technology-
	(286)	(113)	Firms (173)	Intensive Firms
			. ,	(167)
ln(R)	-0.001	0.001	-0.002**	0.000 ^a
	(-1.35)	(0.47)	(-1.81)	(-0.40)
ln(T)	0.002**	0.002	0.001	0.003*
	(1.65)	(1.04)	(0.61)	(2.18)
$\ln(RI^{M}/K)$	0.001	0.001	0.001	0.003*
	(1.21)	(0.84)	(1.05)	(2.14)
$\ln(RI^{D}/K)$	0.003*	0.011*	0.002	0.014*
	(2.24)	(2.08)	(1.36)	(2.75)
Aln(K)/Aln(Q)	0.04*	0.05*	0.04*	0.04*
	(3.51)	(2.78)	(3.31)	(2.67)
Aln(L)/Aln(Q)	0.31*	0.23*	0.36*	0.26*
	(20.78)	(9.34)	(20.57)	(13.02)
Aln(M)/Aln(Q)	0.61*	0.69*	0.59*	0.66*
	(65.89)	(43.43)	(53.65)	(53.11)
Adjusted R ²	0.97	0.97	0.98	0.97

Note: T-statistics are in parenthesis. Output elasticities have been computed at mean values and the associated t-statistics have been computed using the delta method. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Variables	Food	Textiles	Light	Metals	Chemicals	Transport	Electricals	Non-
,	(15)	(48)	(34)	(22)	(80)	(25)	(33)	Electricals
	· · ·		~ /			. ,	~ /	(29)
ln(R)	0.002	-0.001	-0.009*	-0.003	0.000^{a}	-0.002	-0.002	0.005**
	(0.85)	(-0.57)	(-2.14)	(-1.13)	(0.19)	(-0.90)	(-0.87)	(1.73)
ln(T)	-0.006	0.003	-0.001	-0.004	0.003	-0.001	0.008*	0.010*
	(-1.01)	(1.47)	(-0.35)	(-1.47)	(1.61)	(-0.44)	(2.44)	(3.81)
$\ln(RI^{M}/K)$	0.000^{a}	-0.001	0.002	0.001	0.003**	0.010*	-0.002	0.009*
	(-0.23)	(-0.96)	(0.75)	(0.89)	(1.79)	(3.01)	(-0.82)	(2.70)
$\ln(RI^D/K)$	0.016	0.004*	0.006	0.002	0.018**	0.020	0.010	0.003
	(1.39)	(2.01)	(1.36)	(1.18)	(1.89)	(1.60)	(1.50)	(0.12)
ln(K)	-0.11*	0.04*	0.09*	0.02*	0.04	0.08*	0.06*	-0.02
	(-2.02)	(2.35)	(2.55)	(0.89)	(1.54)	(3.04)	(1.92)	(-0.42)
ln(L)	0.35*	0.31*	0.45*	0.27*	0.36*	0.28*	0.20*	0.57*
	(5.92)	(12.51)	(9.48)	(6.62)	(11.00)	(6.82)	(5.89)	(10.48)
ln(M)	0.76*	0.66*	0.43*	0.65*	0.65*	0.64*	0.80*	0.50*
	(23.86)	(35.84)	(22.32)	(23.21)	(62.08)	(23.79)	(35.22)	(20.04)
Adjusted	0.98	0.99	0.97	0.98	0.97	0.99	0.98	0.96
\mathbb{R}^2								

Table 4a Cobb-Douglas production function Fixed effects model with AR1 correction (Time dummies included in all regressions)

Note: T-statistics are in parenthesis. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. ^a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Variablas	Food	Torrtilog	Licht	Matala	Chamicala	Transport	Electricala	Non
variables	гооа	Textiles	Light	Wietais	Chemicals	Transport	Electricals	INOII-
	(15)	(48)	(34)	(22)	(80)	(25)	(33)	Electricals
								(29)
ln(R)	0.003	0.004**	-0.012*	-0.003	-0.001	-0.003	0.002	0.002
	(0.85)	(1.75)	(-3.79)	(-1.31)	(-0.30)	(-1.41)	(0.76)	(0.81)
ln(T)	-0.011	0.005**	-0.001	-0.006*	0.002	-0.002	0.011*	0.010*
	(-1.61)	(1.91)	(-0.48)	(-1.97)	(0.93)	(-0.49)	(2.82)	(3.65)
$\ln(RI^{M}/K)$	0.000^{a}	-0.001	0.001	0.002	0.003	0.010*	-0.004	0.009*
	(-0.13)	(-0.59)	(0.50)	(0.94)	(1.39)	(2.26)	(-1.28)	(2.60)
ln(RI ^D /K)	0.006	0.006**	0.007*	0.004**	0.013	0.034*	0.013**	-0.004
	(0.48)	(1.85)	(2.26)	(1.65)	(1.41)	(1.99)	(1.67)	(-0.20)

Table 4b Pooled Cobb-Douglas production function Fixed effects model with AR1 correction (Time dummies included in all regressions)

Note: T-statistics are in parenthesis. The parameter estimates (and their t-statistics in parenthesis) for K, L, and, M are as follows: (i) Scientific firms: 0.05 (2.75), 0.31 (12.86), 0.67 (72.27); and (ii) Non-Scientific firms: 0.02 (1.85), 0.40 (21.98), 0.53 (52.53). The adjusted R²s for the two groups are 0.98 and 0.97, respectively. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Variables	All Firms	Scientific Firms	Non-Scientific	Technology-
	(286)	(113)	Firms (173)	Intensive Firms
				(167)
ln(R)	-0.001	0.000^{a}	-0.001	-0.002
	(-0.79)	(0.12)	(-0.32)	(-0.71)
$\ln(T)$	0.003**	-0.001	0.003	0.000^{a}
	(1.67)	(-0.29)	(1.48)	(-0.18)
ln(R) ($n(T)$	0.000^{a}	0.000^{a}	0.000^{a}	0.000^{a}
	(0.44)	(0.22)	(-0.36)	(0.58)
$\ln(RI^{M}/K)$	0.000^{a}	0.007**	0.001	0.009*
	(-0.01)	(1.67)	(0.36)	(2.77)
$\ln(RI^D/K)$	0.012*	0.060*	0.011*	0.071*
	(3.96)	(2.99)	(3.59)	(4.11)
$\ln(RI^{M}/K)$ (RI^{D}/K)	0.001*	0.003*	0.001	0.004*
	(2.27)	(2.62)	(1.99)*	(3.65)
$\ln(R)$ (RI^{M}/K)	0.000^{a}	0.000^{a}	0.000^{a}	0.000^{a}
	(0.04)	(-0.50)	(0.34)	(-1.31)
$\ln(R)$ (RI^{D}/K)	0.000^{a}	0.001	0.000^{a}	0.000^{a}
	(0.86)	(0.83)	(0.83)	(0.66)
$\ln(T) \oplus (RI^{M}/K)$	0.0003*	0.000^{a}	0.0003*	0.000^{a}
	(2.92)	(0.34)	(2.77)	(1.49)
$\ln(T)$ $\ln(RI^{D}/K)$	-0.0018*	-0.003*	-0.0004*	-0.003*
	(-2.73)	(-2.85)	(-2.11)	(-3.62)
ln(K)	0.03*	0.04**	0.04*	0.02
	(2.50)	(1.93)	(2.79)	(1.27)
ln(L)	0.35*	0.31*	0.40*	0.32*
× /	(23.36)	(12.84)	(21.92)	(16.07)
ln(M)	0.61*	0.67*	0.53*	0.65*
× /	(87.64)	(72.22)	(52.27)	(79.18)
Adjusted R ²	0.97	0.97	0.97	0.97

Table 5a
Cobb Douglas with flexible functional form for technology stocks
Fixed effects model with AR1 correction
(Time dummies included in all regressions)

Note: T-statistics are in parenthesis. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. ^a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Variables	Food	Textiles	Light	Metals	Chemicals	Transport	Electricals	Non-
	(15)	(48)	(34)	(22)	(80)	(25)	(33)	Electricals
			. ,			. ,		(29)
ln(R)	0.018*	0.007**	-0.014*	-0.007	0.000^{a}	-0.008	-0.018*	0.003
	(2.53)	(1.76)	(-3.56)	(-0.86)	(-0.04)	(-1.59)	(-2.02)	(0.48)
ln(T)	0.107	0.005	-0.005	-0.007**	-0.001	-0.005	0.002	-0.002
	(1.24)	(1.32)	(-0.97)	(-1.72)	(-0.13)	(-0.78)	(0.29)	(-0.36)
ln(R) $an(T)$	-0.006	0.000^{a}	0.000^{a}	0.000^{a}	0.000^{a}	0.000^{a}	0.001*	0.000^{a}
	(-1.53)	(-0.39)	(0.15)	(0.13)	(-0.81)	(0.25)	(2.81)	(0.45)
$\ln(RI^{M}/K)$	-0.006	-0.006	0.008**	0.009**	0.010**	0.022**	0.007	0.013
	(-0.61)	(-1.63)	(1.69)	(1.70)	(1.70)	(1.77)	(0.65)	(1.48)
$\ln(\mathrm{RI}^\mathrm{D}/\mathrm{K})$	0.018	-0.007	0.017*	0.021*	0.063*	0.119*	0.052	0.145**
	(0.20)	(-0.73)	(3.19)	(3.16)	(2.50)	(2.40)	(1.37)	(1.86)
$\ln(RI^{M}/K)@$ $\ln(RI^{D}/K)$	0.000 ^a	-0.003	0.002*	0.003*	0.003	0.007*	0.003	0.009
	(0.08)	(-1.39)	(2.11)	(2.38)	(1.19)	(1.96)	(1.20)	(1.06)
ln(R)@ $ln(RI^M/K)$	0.002*	0.000^{a}	0.000 ^a	-0.001	0.000 ^a	-0.001	-0.001	0.000 ^a
	(3.59)	(0.80)	(-1.53)	(-1.36)	(-0.71)	(-0.78)	(-1.20)	(-0.14)
ln(R)@ ln(RI ^D /K)	-0.002	0.001	0.000 ^a	0.002				
· /	(-0.82)	(0.53)	(-0.91)	(0.91)	(0.30)	(-0.07)	(0.04)	(0.90)
ln(T)@ ln(RI ^M /K)	0.007*	0.000ª	0.000 ^a	0.000 ^a	0.000ª	0.001	0.000 ^a	0.000 ^a
× ,	(1.95)	(0.19)	(0.91)	(0.01)	(0.33)	(0.79)	(0.24)	(0.94)
ln(T)@ ln(RI ^D /K)	0.008	-0.001	-0.005*	-0.001*	-0.004*	-0.004	-0.001	-0.010*
()	(0.35)	(-0.34)	(-3.61)	(-2.04)	(-2.89)	(-1.29)	(-0.29)	(-2.51)

Table 5b
Pooled Cobb Douglas with flexible functional form for technology stocks
Fixed effects model with AR1 correction
(Time dummies included in all regressions)

Note: T-statistics are in parenthesis. The parameter estimates (and their t-statistics in parenthesis) for K, L, and, M are as follows: (i) Scientific firms:0.04 (2.09), 0.32 (12.93), 0.67 (71.55); and (ii) Non-Scientific firms: 0.04 (2.90), 0.39 (21.40), 0.53 (52.11). The adjusted R²s for the two groups are 0.97 and 0.98, respectively. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant.

Variables	All Firms	Scientific Firms	Non-Scientific	Technology-
	(286)	(113)	Firms (173)	Intensive Firms
				(167)
ln(R)	-0.006	-0.003	0.002	0.000^{a}
	(-1.25)	(-0.49)	(0.37)	(0.00)
ln(T)	0.014*	0.009**	0.009	0.009**
	(2.43)	(1.95)	(1.32)	(1.64)
$\ln(RI^{M}/K)$	-0.004	-0.001	0.005	-0.001
	(-1.56)	(-0.20)	(1.50)	(-0.49)
$\ln(RI^D/K)$	0.007*	0.006	0.012*	0.001
	(2.81)	(0.55)	(2.83)	(0.13)
ln(K)	0.02	0.13**	-0.04	0.12*
	(0.34)	(1.81)	(-0.89)	(2.04)
ln(L)	0.47*	0.23*	0.48*	0.25*
	(4.52)	(2.62)	(5.10)	(3.47)
ln(M)	0.52*	0.75*	0.44*	0.73*
	(6.64)	(15.71)	(6.14)	(18.21)
ρ	0.53*	0.40*	0.33*	0.48*
	(8.64)	(4.58)	(4.18)	(6.20)
AR(2) test	-1.20	-1.04	-0.87	-1.27
	(0.23)	(0.30)	(0.38)	(0.20)
Sargan test	241.22	106.17	166.99	160.62
-	(0.12)	(0.32)	(0.22)	(0.34)
Years ^b	1981-87	1982-87	1982-87	1982-87
Instrument lags ^c	3 to 6	3 to 4	3 to 5	3 to 5

Table A1 Cobb-Douglas production function GMM-IV on first differenced data (Time dummies incorporated in all regressions)

Note: Numbers in parenthesis are t-statistics except in case of tests for AR2 and over-identifying restrictions (Sargan test) where they correspond to P-values. * Significant at 0.05 level in a two-tailed test. ** Significant at 0.10 level in a two-tailed test. ^a The parameter estimate is 0 when rounded to the third decimal place and statistically insignificant. ^{b, c} The use of lagged variables as instruments necessitates sacrificing initial years of data. Additionally, because the degrees of freedom vary with the number of observations, different lag lengths can be used across the four sets of estimates. The procedure used here is to choose longer lag lengths where possible so as to improve efficiency.